Author’s response to reviews

Title: Implementation of a Successful Eradication Protocol for Burkholderia Cepacia Complex in Cystic Fibrosis Patients

Authors:

Bryan Garcia (bryangarcia25@gmail.com)
Jacque Carden (jlcarden@uabmc.edu)
Dana Goodwin (dgoodwin@uabmc.edu)
Tim Smith (tasmith@uabmc.edu)
Amit Gaggar (agaggar@uabmc.edu)
Kevin Leon (kleon@uabmc.edu)
Veena Antony (Vantony@uabmc.edu)
Steven Rowe (srowe@uabmc.edu)
George Solomon (gsolomon@uabmc.edu)

Version: 2 Date: 22 Jan 2018

Author’s response to reviews:

On behalf of the co-authors and myself, we have updated a revised version of the manuscript with the recommended minor grammatical changes based on the editors recommendations and this has been attached as the manuscript.